Author’s response to reviews

Title: Curbing the lifestyle disease pandemic: making progress on an interdisciplinary research agenda for law and policy interventions

Authors:

Brigit Toebes (b.c.a.toebes@rug.nl)
Marlies Hesselman (m.m.e.hesselman@rug.nl)
Jitse Van Dijk (j.p.van.dijk@umcg.nl)
Joost Herman (j.herman@rug.nl)

Version: 1 Date: 31 Jul 2017

Author’s response to reviews:

Groningen, 26 July 2017

Dear Editors,

Thank you for your comments and the opportunity to revise the manuscript. In our 'List of changes' document you will find an overview of changes relating to the manuscript IHHR-D-17-00040 entitled “Curbing the lifestyle disease pandemic: a multidisciplinary research agenda for law and policy interventions”.

We provide a point-by-point response, with each response preceded by the reviewer’s comment. The changes made in the manuscript are highlighted in track changes. Thus, all changes can be read in the manuscript. The changes can also be found in italics the present document.

On behalf of the co-authors,

Brigit Toebes
Professor Brigit Toebes
Associate Professor and Rosalind Franklin Fellow
T +31(0)503635676
University of Groningen
Faculty of Law, International Law
Global Health Law Groningen: www.rug.nl/law/ghlg
Email b.c.a.toebes@rug.nl